
Daily Status Saturday 10-27-12
Weather: 

 A little bit of cloud cover, calm winds through out the night. Temps around 3 to 5 C.

Status:

All systems looked good throughout the night except for 10m krypton which is still out. I will try to get up the tower if the RAL person is still here.

The power status at station 6 down. Gordon did a power cycle earlier today which brought it back but it has gone down again, will investigate.

Qsoilc is down.

UPDATE:

Went up the tower to investigate the bad krypton  (RAL person still here). Decided to disconnect/connect all cables to krypton. Did not help.

Went to station 6 to look at Vdsm issue. Power cable to mote box was a little loose, reseated and recycled power, no improvement.

Cleaned radiometer domes with methanol. Reseated connectors at soil cactus site. Qsoilc came back.
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